Author’s response to reviews

Title: A mutation in the filamin c gene causes myofibrillar myopathy with lower motor neuron syndrome: a case report

Authors:
Juanjuan Chen (chenjuanjuan_1025@163.com)
Jun Wu (wujun188@163.com)
ChunXi Han (cxhan60@126.com)
Yao Li (liyao2115918@163.com)
Yuzu Guo (759838694@qq.com)
Xiaoxin Tong (tongxiaoxin100@foxmail.com)

Version: 4 Date: 19 Jul 2019

Author’s response to reviews:

Dear Editors,

Thank you for accepting our manuscript entitled "A mutation in the filamin c gene causes myofibrillar myopathy with lower motor neuron syndrome: a case report" (NURL-D-18-00917R1).

The corrections in the paper and the responds to the Editor comments are as following:

Responds to the Editor’s comments: It is noticed that you have written in the Ethics approval and consent to participate subsection, as well as the Consent for publication subsection that the “healthy unrelated controls” provided written informed consent for participation and publication for this study. Can you please clarify whether you actually meant that you approached each individual control for the consent pertaining to this study, or that you compared the current patient’s test results to that of the retrospective data of the controls?
We compared the current patient’s test results to that of the retrospective data of the controls. So we revise our manuscript Line 277-278 as “Not applicable. Written informed consent for participating in this study was obtained from the patient.”

Thank you very much for your comments.

Best wishes,

Jun Wu

wujun188@163.com